Author’s response to reviews

Title: A Gaze-independent Audiovisual Brain-Computer Interface for Detecting Awareness of Patients with Disorders of Consciousness

Authors:
Jiahui Pan (panjh82@qq.com)
Jiahui Pan (panjh82@qq.com)
Yan Chen (269837537@qq.com)
Yanbin He (17336071@qq.com)
Xiaoxiao Ni (862121021@qq.com)
Jiechun Zhang (353336293@qq.com)
Fei Wang (sandra88@qq.com)
Yuanqing Li (auyqli@scut.edu.cn)
Ronghao Yu (gesund@21cn.com)

Version: 4 Date: 21 Jul 2018

Author’s response to reviews:

Responses to the Editor

The authors are highly grateful to the editors for coordinating the review of our manuscript. In light of your comments and suggestions, the paper has been revised. Please see our point to point responses in the following.

Technical Comments:

Editor Comments:

Reviewer reports:
If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication.

Response: According to your comment, we have improved the English as your suggestion. Specifically, we sent this manuscript to a company American Journal Experts (a professional editing website, https://www.aje.com/) for improving the English.